## UC Santa Barbara

**UC Santa Barbara Previously Published Works** 

## Title

Selection on modifiers of genetic architecture under migration load.

## Permalink

https://escholarship.org/uc/item/2rk1b7dq

## **Journal** PLoS Genetics, 18(9)

**Authors** Proulx, Stephen Teotónio, Henrique

## **Publication Date**

2022-09-01

## DOI

10.1371/journal.pgen.1010350

Peer reviewed



## G OPEN ACCESS

**Citation:** Proulx SR, Teotónio H (2022) Selection on modifiers of genetic architecture under migration load. PLoS Genet 18(9): e1010350. https://doi.org/10.1371/journal.pgen.1010350

**Editor:** Lindi Wahl, University of Western Ontario, CANADA

Received: January 13, 2022

Accepted: July 20, 2022

Published: September 7, 2022

**Copyright:** © 2022 Proulx, Teotónio. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: Data/code are available on data dryad: <u>https://doi.org/10.25349/</u>D9991T.

**Funding:** HT is supported by the Agence Nationale pour la Recherche (ANR-17-CE02-0017-01, ANR-18-CE02-0017-01) Include this sentence at the end of your statement: The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

**Competing interests:** The authors have declared that no competing interests exist.

**RESEARCH ARTICLE** 

# Selection on modifiers of genetic architecture under migration load

#### Stephen R. Proulx<sup>1\*</sup>, Henrique Teotónio<sup>2</sup>

1 Department of Ecology, Evolution, and Marine Biology, UC Santa Barbara, Santa Barbara, California, United States of America, 2 Institut de Biologie, École Normale Supérieure, Paris, France

\* stephen.proulx@gmail.com

## Abstract

Gene flow between populations adapting to differing local environmental conditions might be costly because individuals can disperse to habitats where their survival is low or because they can reproduce with locally maladapted individuals. The amount by which the mean relative population fitness is kept below one creates an opportunity for modifiers of the genetic architecture to spread due to selection. Prior work that separately considered modifiers changing dispersal, recombination rates, or altering dominance or epistasis, has typically focused on the direction of selection rather than its absolute magnitude. We here develop methods to determine the strength of selection on modifiers of the genetic architecture, including modifiers of the dispersal rate, in populations that have previously evolved local adaptation. We consider scenarios with up to five loci contributing to local adaptation and derive a new model for the deterministic spread of modifiers. We find that selection for modifiers of epistasis and dominance is stronger than selection for decreased recombination, and that selection for partial reductions in recombination are extremely weak, regardless of the number of loci contributing to local adaptation. The spread of modifiers that reduce dispersal depends on the number of loci, epistasis and extent of local adaptation in the ancestral population. We identify a novel effect, that modifiers of dominance are more strongly selected when they are unlinked to the locus that they modify. These findings help explain population differentiation and reproductive isolation and provide a benchmark to compare selection on modifiers under finite population sizes and demographic stochasticity.

### Author Summary

When populations of a species are spread over different habitats the populations can adapt to their local conditions, provided dispersal between habitats is low enough. Natural selection allows the populations to maintain local adaptation, but dispersal and gene flow create a cost called the migration load. The migration load measures how much fitness is lost because of dispersal between different habitats, and also creates an opportunity for selection to act on the arrangement and interaction between genes that are involved in local adaptation. Modifier genes can spread in these linked populations and cause functional, local adaptation genes, to become more closely linked on a chromosome, or change the way that these genes are expressed so that the locally adapted gene copy becomes dominant. We modeled this process and found that selection on modifiers that create tighter linkage between locally adapted genes is generally weak, and modifiers that cause gene interactions are more strongly selected. Even after these gene interactions have begun to evolve, further selection for increased gene interaction is still strong. Our results show that populations are more likely to adapt to local conditions by evolving new gene interactions than by evolving tightly linked gene clusters.

#### Introduction

Gene flow is a fundamental process that introduces, maintains, or reduces genetic variation between demes of a population adapted to local environmental conditions. Along with this variability comes a fitness cost, the "migration load", that decreases the population mean fitness below that of a deme lacking immigration [1-5]. It has long been known that selection in natural populations can act to reduce the dispersal or recombination rates between loci experiencing recurrent mutation, dubbed the "reduction principle", cf. [6]. Much of the study on the reduction principle has been on proving that the direction of selection is towards reducing mixing of genotypes, rather than specifying the relative magnitude of selection on modifiers of the genetic architecture that can, in principle, diminish migration loads. Heritability for these modifiers has been described in the empirical literature including those that change dispersal rates [7], recombination rates [8, 9], and those that alter dominance [10, 11] and epistasis [12, 13] within and between locally adapted loci. Many of these types of modifiers may segregate concurrently within and between demes during and after the process of local adaptation. It is thus important to quantify the relative magnitude of potential selection on modifiers of genetic architecture to understand differentiation between populations, stability of hybrid zones and speciation.

Interest about selection on modifiers of genetic architecture under migration load has been spurred by the observation that genomic regions contributing to phenotypic differentiation and reproductive isolation between populations or incipient species do not often recombine [14-17]. How these "genomic islands" of differentiation and isolation originate and are maintained is intriguing. One possibility is that selection causes a build up of linked alleles that contribute to local adaptation, either by favoring mutations, such as chromosomal inversions, that eliminate recombination between pre-existing locally-adapted alleles [18-20], or by favoring mutations that just happen to be linked to locally-adapted alleles when they appear during ongoing local adaptation [21-27]. Another possibility is that genomic islands of differentiation and isolation are created by genetic drift and the accumulation of mutations in genome locations of reduced recombination rates under allopatry, in a manner similar to how Dobzhansky-Muller incompatibilities arise and are maintained [28, 29].

Most theoretical work so far has described the expected direction and strength of selection on recombination modifiers under several simplifying assumptions [18–20, 30]. First, the recombination modifier is usually modeled as a chromosomal inversion so that its spread can be calculated without taking into account recombination of the modifier into alternative genotypes. Second, the fitness functions are generally assumed to be additive, meaning there is no dominance between locally-adapted alleles or epistasis between locally-adapted loci. Third, migration is usually assumed to be "continent to island" and thus dispersal is uni-directional. Under weak migration and no epistasis between loci, for example, the strength of selection on an inversion depends on the migration load, and that selection increases as the number of locally-adapted loci increase and is higher when the ancestral genotype has free recombination [20]. When the modifier creates chromosomal inversions, the strength of selection on it is completely determined by the fitness load present in the structured population [31].

A separate literature has considered the evolution of dominance modifiers, dating back to early work by Fisher [32] and Wright [33]. A major objection to Fisher's notion that dominance modifiers could be selected to favor wild-type alleles under mutation-selection balance is that heterozygotes are rare and dominance modifiers are therefore only weekly selected [34– 36]. However, if other processes maintain heterozygotes at high frequencies, then selection for dominance modifiers can be substantial [37, 38]. For example, when there is dispersal between demes facing different environmental conditions, migration-selection balance can lead to the long-term maintenance of heterozygotes and result in selection for dominance modifiers [37]. Understanding selection on modifiers of epistasis is complex, depending on the sign of linkage disequilibrium between deleterious alleles being purged or beneficial alleles being favored with the mutant modifier alleles, as well as on the dominance interactions at each locus [27, 39]. We do know, however, that in general an epistatic modifier affecting the total fitness load of a structured population is expected to be strongly selected relative to modifiers affecting only specific loci [31].

When there is multi-directional dispersal between demes, calculation of relative selection strength on modifiers of genetic architecture depends on the full distribution of genotype densities in the demes. Much of the discussion has centered on the distribution of offspring genotypes of a given parent, but this view obscures the fact that modifiers alter their own associations with alternative genotypes, and therefore that the strength of selection depends both on the frequency of genotype or environment backgrounds in the absence of the modifier as well as on the reproductive value of those states. For example, [40] explore this interaction when alleles affecting mating preferences are selected under migration-selection balance. They find that selection on changes in the transition probability from a particular genotype or environmental background will be weak if the frequency of that state in the stable stage distribution (absent the modifier) is low, even if the potential fitness benefit of a transmission modification is high. Conversely, selection on changes in the transition probability from a genotype or environment background will be weak if the absolute reproductive value of the new background state (absent the modifier) is low.

Selection on modifiers of the genetic architecture under migration load will be strongest for changes in the transition probabilities that affect relatively frequent states and cause a large change in the reproductive value of their descendant states, possibly leading to local extinction [41]. Because modifiers affect these transition probabilities without altering the total number of offspring produced, any increase in the probability of transition between two states must be associated with an equal reduction of transmission to other states. Therefore, one must calculate the cumulative effect of these changes in transmission in order to quantify the magnitude of selection on modifiers of genetic architecture under migration load.

We here investigate selection in a scenario with bi-directional dispersal between two demes that face two habitats with differing environmental conditions, where a diploid genotype is determined by a linear chromosome with arbitrary recombination between several adjacent loci (up to five), and where fitness can include dominance and epistasis. Although we assume a linear chromosomal recombination map, if the recombination probability is 0.5, then we recover a scenario with freely segregating chromosomes. Mating between individuals, precisely syngamy, is assumed to be random. We numerically solve for the equilibrium—a steady-state between migration, selection, segregation, recombination and syngamy—of genotype frequencies in each deme and calculate the migration load. We then introduce a rare mutant modifier of the genetic architecture and calculate the selection strength on the invading modifier as the eigenvalue of the transition matrix. The modifier can independently alter the overall magnitude of the migration load, the dominance at single or multiple loci, the epistasis between loci, the recombination rate between loci, and the migration rate between demes.

#### Methods

#### Life-cycle

We consider two demes with reciprocal migration, each adapting to habitats with different environmental conditions. After birth, diploid individuals migrate between the two demes and are subject to viability selection within each deme (see next section for the description of the fitness function). After selection, gametes are produced by meiosis with Mendelian segregation and cross-over recombination between homologous chromosomes. Recombination is modeled as if there were a genetic linkage map of a single linear chromosome. The locations of the k loci are given by their index and a recombination vector describes the probability of crossing-over between successive loci, giving k - 1 recombination rates. After recombination, gametes are produced by defining a table of the frequency of all possible gamete types given each possible pair of parental haplotypes. Following [42], a square matrix G with dimensions  $2^k$  is defined such that each element  $G_{i, i}$  is itself a vector of length  $2^k$  (technically G is a tensor of 3 dimensions). Each position *l* in the vector represents the probability that a parent with haplotypes *i* and *j* would produce a gamete with index *l*. The gamete production table is computed for situations with up to k = 5 local adaptation loci and 1 modifier loci. This model has a total of 6 loci with 64 haplotypes, and thus, the gamete production table has  $64^3 = 262$ , 144 entries. These gamete tables are defined generically without inserting numerical values and stored as an object. Values can be efficiently substituted into these stored objects, meaning that the gamete table does not need to be re-calculated during simulations, but it does need to be evaluated with the recombination parameter values once per simulation of the modifier invasion (see below).

Variable	Definition
k	The number of genes contributing to local adaptation
$l_i(h_1, h_2)$	The effective number of locally adapted alleles
S	Is the total fitness cost associated with the least-adapted genotype
$m_d$	The migration rate from deme <i>d</i>
$r_m$	The recombination rate for the modifier
r <sub>i</sub>	The recombination frequency between locus $i$ and $i + 1$
$\phi$	The epistasis parameter in favor of the locally adapted allele in deme 1
$\theta_i$	The dominance parameter for locus <i>i</i> in favor of the locally adpated allele in deme 1
G	The matrix storing the probability of producing each possible gamete haplotype for each diploid genotype.
$w_{i,j}^d$	The fitness of an adult carrying haplotypes <i>i</i> and <i>j</i> in deme <i>d</i> .
$\overline{w}^{d}$	The mean fitness in deme d
$\overline{x_i^d}$	The frequency of haplotype <i>i</i> in deme <i>d</i> .
$\hat{z}$	The parameters associated with the modifier allele are indicated with ^
A	The invasion matrix for the mutant modifier allele.
λ	The dominant eigenvalue of <b>A</b>
L <sub>d</sub>	The fitness load in deme <i>d</i>

Table 1. Notation.

The life cycle just described can be concisely expressed as:

$$x_{i}^{d\prime} = \sum_{j} \sum_{k} w_{j,k}^{d} \left( x_{j}^{d} x_{k}^{d} (1 - m_{d}) + \sum_{\widehat{d} \neq d} x_{j}^{\widehat{d}} x_{k}^{\widehat{d}} m_{\widehat{d}} \right) G_{i,j,k}$$
(1)

$$\bar{w^d} = \sum_i x_i^{d\prime} \tag{2}$$

$$x_i^{d\prime\prime} = \frac{x_i^{d\prime}}{\overline{w^d}},\tag{3}$$

where  $x_i^d$  is the frequency of haplotype *i* in deme *d*,  $w_{j,k}^d$  is the fitness of genotype (j, k) in deme *d*,  $G_{i,j,k}$  represents the probability that an adult with haplotypes *j* and *k* will produce a gamete of haplotype *i*, and  $\overline{w^d}$  is the average fitness within deme *d*. Haplotype frequency in the next generation is given by  $x_i^{d''}$ . See the Table 1 for a full list of symbols used.

We also have developed a notation system to describe the life-cycle and the dynamics of modifiers using linear algebra operators, and this is detailed in Appendix A in <u>S1 Appendix</u>. The main benefit of the linear operator notation is that it separates the life cycle into modules that can be easily modified or amended, for instance to include non-random mating, parental effects, or non-Mendelian segregation.

#### **Fitness function**

Diploid individual fitness is a function of both habitat and genotype. We model fitness by assuming that each locus has a parameter determining dominance between alleles at each of the *k* local adaptation loci, and that epistasis between all *k* local adaptation loci are determined by a single parameter. Our approach is to use generalized logistic functions to describe the *effective* number of locally adapted alleles. Specifically, for a population with *k* loci, each individual has 2k alleles that are each either locally adapted (allele state of 1 in deme 1) or not (allele state of 0 in deme 1). Fitness in habitat *d* for an individual that inherited haplotypes  $h_1$  and  $h_2$  is then:

$$W_d(h_1, h_2) = e^{l_d(h_1, h_2)\log(1-S)},$$
(4)

where *S* represents the total fitness cost of having the most maladapted genotype for the resident haplotypes, and  $l_d(h_1, h_2)$  represents the effective fraction of alleles that are maladapted carried by an  $(h_1, h_2)$  individual in habitat *d*. In our parameterization, when the dominance and epistasis parameters are set to 0,  $l_d(h_1, h_2)$  is equal to the actual fraction of alleles that are maladapted. In other words, when there is no epistasis or dominance the effective number of locally adapted alleles is the same as the actual number of locally-adapted alleles. If  $l_d = 1$ , then fitness is 1 - S, and if  $l_d = 0$  then fitness is 1.

For dominance, each locus is parameterized by  $\theta_i$ , which determines how the alleles interact. At a locus *i*, the effective fraction of locally adapted alleles is 0 if the individual is homozygous for the maladapted allele, 1 if the individual is homozygous for the locally adapted allele. If the individual is heterozygous then locus specific fitness is  $w_i = 1 - h_i s$  where

$$h_i = \frac{1}{1 + \exp(-\theta_i)}.$$
(5)

This gives  $h_i = 1/2$  if  $\theta_i = 0$ , indicating that if there is no dominance in either direction, and

heterozygotes behave as if 1/2 of their alleles are locally adapted. In the limit as  $\theta_i \to \infty$  the fraction adapted goes to 1, and as  $\theta_i \to -\infty$  the fraction adapted goes to 0.

Once we have calculated dominance at each individual locus, we sum across loci to determine an aggregate effect:

$$dom(h_1, h_2) = \frac{1}{k} \sum_{i=1}^k \left( \frac{\operatorname{het}_i}{1 + \exp(-\theta_i)} + \operatorname{hom}_i \right)$$
(6)

with het<sub>*i*</sub> an indicator variable that takes on 1 if the individual is heterozygous at locus *i* and 0 otherwise, hom<sub>*i*</sub> an indicator variable that takes on the value 0 if the individual is homozygous for the locally favored allele at locus *i* and 1 otherwise.

Epistasis is applied in a similar way as dominance except now the fraction of adapted alleles is calculated across loci:

$$l(h_1, h_2) = \frac{1}{1 + \exp\left(\phi - \log\left(\frac{dom(h_1, h_2)}{1 - dom(h_1, h_2)}\right)\right)},$$
(7)

where  $\phi$  is the epistasis parameter. When  $\phi = 0$ , then  $l(h_1, h_2) = dom(h_1, h_2)$ , so that the fraction of adapted alleles is unaltered by epistasis. The epistasis calculation depends on both  $\phi$  and the fraction of adapted alleles determined by dominance. If  $dom(h_1, h_2) = 0$ , then l = 1 for any finite value of  $\phi$ . Likewise, if  $dom(h_1, h_2) = 1$ , then l = 0 for any finite value of  $\phi$ . If  $0 < dom(h_1, h_2) < 1$  then as  $\phi \to \infty l$  approaches 0 while as  $\phi \to -\infty l$  approaches 1.

Under this formulation, fitness is additive when  $\phi = 0$  and  $\theta_i = 0$  (for all *i*). Further, changes in the  $\theta$  and  $\phi$  from 0 have symmetric effects on the log scale. As  $\phi$  and  $\theta$  move away from 0, the effect quickly saturates (Fig 1A). For dominance, if gene regulation is environment dependent, for example, one might find that a single modifier allele causes locally adapted alleles to become dominant in the habitats they perform best in and conversely more recessive in the habitats they perform less well (Fig 1B). This is reminiscent of classical physiological explanations of partial dominance [43]. Similarly for epistasis [39], adding locally adapted homozygous loci has diminishing fitness returns in habitats where the population performs well, while it is synergistic in habitats where the population performs less well (Fig 1C).



**Fig 1. The fitness function showing the effects of dominance and epistasis.** The total fitness cost is set at S = 0.1. Panel A shows the effect of changing the dominance and epistasis parameters in a genotype composed of 20 loci, 5 of which are homozygous for the adapted allele, and 10 of which are homozygous for the maladapted allele. The black line is the fitness value with no dominance or epistasis shown for reference. Note that in this example  $\theta$  is the same for all loci. In panel B, the effect of dominance on fitness is shown for a single locus, as the number of adapted alleles is varied from 0 (homozygote) to 1 (heterozygote) to 2 (homozygote). The effect is shown for two values of the dominance parameter  $\theta$ . Panel C shows the effect of the epistasis parameter in a genotype with 20 loci, as the number of loci homozygous for the adapted allele is varied. Two values of the epistasis parameter,  $\phi$ , are shown.

#### Ancestral population

We use iterations of the recursion equations shown above for the life-cycle to obtain a population of "resident" haplotypes at an equilibrium between standing variation, dispersal, recombination, mating and selection. We initialize the recursion equations with the locally favored haplotype being common (frequency of 0.9999) and with the other haplotype frequencies assigned at random. This avoids the pitfall of having the simulation maintain a symmetric but unstable equilibrium. We find equilibrium haplotype frequencies by iterating the recursion equations while censusing the population at the haploid gametic stage. Iteration occurs for a maximum of 5000 generations or until the haplotype frequencies change by less than  $10^{-8}$  (see supplemental archived Mathematica file for further details [44]).

#### Selection on the modifier mutant allele

The primary goal of our study is to calculate the magnitude of the leading eigenvalue of an invasion matrix describing modifiers of the genetic architecture of the resident population. The genetic architecture is fixed in the resident ancestral population, which means, for example, that the recombination rates between loci do not depend on the allele effects at those loci. Local adaptation can be maintained so long as dispersal is not too high and when selection for locally adapted alleles is relatively balanced among the habitats. Even when there is initially no dominance or epistasis determining fitness, heterozygosity is maintained at migration-selection balance as well as linkage disequilibrium between loci within and between demes. Implicitly, indirect selection on the modifier allele is created by linkage between the invading modifier allele and locally-adapted haplotypes and thus depends on the extent of heterozygosity in the resident population. In all of the analyses presented here we assume symmetric migration rates and selection coefficients, as this ensures that the fully polymorphic equilibrium is stable [30]. Our method works under asymmetry as well, but it then requires additional checking to ensure that a polymorphism in local adaptation is maintained.

The invasion matrix for the mutant modifier allele, **A**, defines the number and genotype of mutant offspring produced per mutant individual in the prior generation. The construction of this matrix follows the same life-cycle as for the resident population, but differs in that the frequency of potential mates is set by the frequency of resident genotypes at the resident equilibrium, and of course in that the mutant parameters are applied. Thus, the matrix elements  $A_{i,j}$  represent the number of diploid adults in the following generation that carry the modifier and the mutant haplotype and resident haplotype denoted by the index *j* that descend from a single individual with the mutant haplotype and resident haplotype denoted by the index *i*.

The modifier is always placed in position 1 of the genetic linkage map. Each pair of adjacent loci experiences an independent chance of crossing-over given by the recombination vector. The modifier may alter the parameters which are indicated by  $\hat{S}$ ,  $\hat{r}_i$  (recombination modifier),  $\hat{\phi}$  (epistasis modifier),  $\hat{\theta}_i$  (dominance modifier), and  $\hat{m}$  (dispersal modifier). Here, we consider modifier alleles that alter only one parameter at a time, although in principle the modifier may alter several or all the parameters at once. Because the mutant modifier allele is rare, matings between two individuals carrying it are even rarer. The recombination mutant modifier allele is dominant, all other mutant modifier alleles are co-dominant.

We start by indexing the vector describing the density of mutants as v, which has dimension  $2^{2^{k+1}}$ . This dimensionality comes from enumerating all combinations of mutant and resident haplotypes in both demes  $(2^{k*}2^{k*}2)$ . The matrix **A** is then constructed in three stages, first with a matrix **D** combining the segregation and recombination matrix from the two demes and then applying migration, and third applying the matrix **F** to adults after migration. The matrix **R**<sup>*j*</sup> has dimension  $2^k$ , and details how segregation, recombination and syngamy produce

offspring in the next generation. If we define the component haplotypes of genotype *i* as  $i_{\mu}$  and  $i_r$  for the mutant and resident haplotypes found in the diploid with index value *i*, then:

$$\mathbf{R}^{k}[i,j] = G_{i_{u},i_{u},i_{r}} * V_{i_{r}} * 2$$

where  $G_{j_{\mu},i_{\mu},i_{r}}$  represents the probability that a gamete with haplotype  $j_{\mu}$  is produced by a diploid bearing haplotypes  $i_{\mu}$  and  $i_{r}$ . This is multiplied by  $v_{j_{r}}$  to represent the probability that the gamete fuses with a resident gamete bearing haplotype  $j_{r}$ . The factor of 2 accounts for the fact that, at population dynamic equilibrium, each diploid adult produces an average of two successful gametes.

Migration then moves mutant genotypes between the two demes. For this, we create a block matrix composed of the recombination/syngamy matrices:

$$\mathbf{D} = \begin{bmatrix} \mathbf{R}^{1} * (1 - \hat{m_{1}}) & \mathbf{R}^{1} * (\hat{m_{1}}) \\ \mathbf{R}^{2} * (\hat{m_{2}}) & \mathbf{R}^{2} * (1 - \hat{m_{2}}) \end{bmatrix}$$

which has dimension  $2^{2k+1}$ .

After migration, we apply selection by creating a vector f of relative fitness values, for each diploid genotype. This vector has length  $2^{2k+1}$  and its elements are:

$$f_i = egin{cases} rac{w_1(\mu_i, R_i)(\hat{S}, \hat{\phi}, \hat{ heta})}{ar{w}_1} & i \leq 2^k \ rac{w_2(\mu_i, R_i)(\hat{S}, \hat{\phi}, \hat{ heta})}{ar{w}_2} & i > 2^k \end{cases}$$

where the fitness function uses the mutant fitness parameters. Because the fitness function is applied at the adult stage, the transmission of genotypes in **D** to the next stage depends only on their column index, not their row index. We define a matrix  $F_{i,i} = f_i$ .

The transition matrix for the invasion modifier is then

$$\mathbf{A} = \mathbf{D} \circ \mathbf{F}$$

The invasion recursion for the modifier across generations t can be described by

$$v_{t+1} = v_t \cdot \mathbf{A}$$

The selection coefficient of the mutant modifier allele, or long-term growth rate multiplier, is the dominant eigenvalue of **A**. With our approach, the potential loss by genetic drift of the mutant modifier is not modelled and thus we assume that once the modifier appears it will be established in the resident population and follow deterministic dynamics [45]. However, the invasion matrix generated here can be used to define a multi-type branching process to study stochastic establishment and will be considered in a future work.

#### **Results and discussion**

#### Invading the ancestral population

We first examine selection on modifiers of the genetic architecture of an ancestral population characterized by multiplicative fitness and free recombination (i.e.  $r_i = 0.5$ ,  $\phi_j = 0$ , and  $\theta_{i,j} = 0$ ). We consider modifiers that alter one aspect of genetic architecture at a time (i.e. that change one of the fitness/recombination/migration parameters). We also assume that modifiers have symmetric effects in the two demes, such that  $\phi_2 = -\phi_1$  and  $\theta_{i,2} = -\theta_{i,1}$ . Biologically, this amounts to assuming that gene expression is environment independent and that fitness effects

are opposite but symmetric between loci [46, 47]. Other assumptions are plausible, and may even be expected when gene expression is modified in an environmentally plastic fashion. Here we stick with this simple assumption of opposite effects and note that this is a conservative assumption with respect to the strength of selection on dominance modifiers. We set relatively high levels of prior local adaptation, with the total fitness cost of the most maladapted genotype being 0.1 and migration rates of 0.05. We further modeled complete linkage between the modifier locus and the first local adaptation locus ( $r_m = 0$ ) and identical values for effect of a modifier on dominance at each locus  $(\hat{\theta}_{i,1})$  and recombination parameter  $(\hat{r}_i)$ . In all figures, modifier parameters are expressed relative to the maximum change possible for that parameter. For example, for recombination,  $\hat{r}$  can be reduced from 0.5 down to 0, so we use the transformation  $(0.5 - \hat{r})/0.5$  to get the fractional change. For dominance and epistasis, the effect of negative values is completely symmetric to that of positive values, representing the opposite polarity for the deme in which the locally adapted allele is dominant. The values of  $\phi$  and  $\theta$  are limited to a maximum of 3 because the effect of the modifier on the effective number of locally adapted alleles is saturated by this point (Fig 1). We used this maximum value of 3 to rescale the dominance and epistasis parameters for the figures.

Results of this analysis are shown in Fig 2. We plot the theoretical maximum for a modifier that does not directly alter fitness, which is strictly a function of the migration load [31]. We also plot, for reference, the magnitude of selection on a modifier that reduces the total negative effect of maladapation, i.e. a modifier that reduces *S* directly. We find that modifiers of recombination rate have the smallest selection coefficients, and that there is very weak selection unless they almost completely eliminate recombination between locally adapted alleles. While selection for reduced recombination is slightly higher when more loci contribute to local adaptation, it is still the case that selection only becomes noticeably strong as recombination gets close to zero.



**Fig 2. Initial strength of selection on different modifiers as measured by**  $\lambda$  – **1.** Parameter values for the ancestral population are *S* = 0.1, *M* = 0.05, *R* = 0.5;  $\theta$  = 0;  $\phi$  = 0. We plot the fraction of the maximum allowable change in the parameter on the x-axis. The dashed horizontal line represents the theoretical maximum for the spread of a modifier, while the solid black line shows the strength of selection for a modifier that reduces the total effect of the maladaptation (i.e. a modifier that reduces *S*). Top panels left to right show results for two local adaptation loci, while bottom panels left to right show results for five local adaptation loci. For dominance modifiers, the modifier is assumed to affect dominance at all local adaptation loci simultaneously. Likewise, the recombination modifier is assumed to reduce recombination between all loci simultaneously. We only show positive changes in dominance and epistasis, because negative values have identical effects but indicate that deme 2's locally adapted allele is instead dominant.

Selection for reduced dispersal into unfavorable habitats is generally stronger than selection for reduced recombination. The strength of selection on migration modifiers decreases substantially as the number of loci increase, even though the magnitude of the migration load is slightly higher. This is because the migration modifier is directly linked to only a single local adaptation locus and can recombine into a more favorable genetic background after migrating into an unfavorable habitat.

Selection is strong on both epistasis and dominance, and saturates as the modifier parameter increases. Even small increases in the dominance or epistasis parameter results in an appreciable selection coefficient. Selection on epistasis is stronger than selection on dominance, for the same level of phenotypic bias, and this difference becomes more noticeable as more loci contribute to local adaptation. Overall, selection for modifiers of epistasis and dominance is stronger than selection for modifiers of recombination and dispersal.

#### The number of local adaptation loci

We next varied the number of loci contributing to local adaptation while keeping the magnitude of dispersal and the migration load constant. More than 5 loci contributing to local adaptation becomes computationally straining, though asymptotic patterns can be observed as we go up to 5 loci. We determine the maximum possible strength of selection when one of the parameters is set to an extreme value. For recombination and dispersal modifiers this involves setting the parameter to 0. For epistasis and dominance modifiers it involves setting parameters to 10 because this causes complete dominance or epistasis.

Results show that as the number of local adaptation loci increases from 2 to 5 the total migration load goes up by about 10%, which increases the maximum potential strength of selection (Fig 3A). Selection on epistasis and recombination modifiers show increases of similar amounts, indicating that the fraction of the migration load that they can ameliorate remains nearly constant. Selection on dominance decreases only slightly with increase number of loci. Most different is the response of dispersal, where the strength of selection decreases by about 50%.

We also evaluated the effect of fitness loci on the strength of selection for moderate effect mutant modifier alleles. For dispersal and recombination modifiers we reduce their parameter values to 50% of the ancestral population rate. For epistasis and dominance we set the parameter to 1.1, because this leads to a 50% reduction in the fitness cost when half the alleles are





locally adapted. We find the pattern of change is largely the same, with the main difference being that the strength of selection on the recombination modifier is relatively smaller than on the other modifier types (Fig 3B).

#### **Dominance modifiers**

In Fig 2 results are presented for modifiers of dominance that affect all fitness loci simultaneously. It is thus interesting to consider modifiers of dominance that affect only one of the local adaptation loci. For this, the modifier is assumed to be at one end of a linear chromosome and completely linked (r = 0) with the first local adaptation locus (called locus 1). In Fig 4 we show the analysis for a scenario where the ancestral population has free recombination (r = 0.5) between successive fitness loci. When the modifier affects locus 1, the strength of selection is lower than when the modifier affects the second locus 2 or a higher index locus in scenarios with more than two loci.

As shown before in Fig 3, selection on modifiers affecting all fitness loci simultaneously decreases slightly as the number of loci increases. When considering modifiers that affect dominance at single local adaptation loci, we find that the strength of selection is substantially lower, and that the effect of modifying multiple loci is more than additive. Regardless of the number of loci, the strength of selection is stronger when the modifier affects unlinked genes. Furthermore, when only two loci contribute to local adaptation the effect of position is much larger than when five loci contribute to local adaptation.

Selection on dominance modifiers depends on two components, the frequency of heterozygotes in the ancestral population and the change in relative fitness that is caused by the modifier. If there is little potential effect of the modifier, i.e. because the cost of having the maladapted allele is low, then selection on the modifier will be weak. Likewise, if heterozygotes are at low frequency, i.e. because there is little migration and strong selection against mutants, then selection on the modifier will be weak. Neither factor alone is particularly predictive of modifier selection strength. For example, if we compare scenarios with different values of *S*, increases in *S* cause heterozygosity to decrease but cause selection on the modifier to increase (not shown).



**Fig 4. Strength of selection on dominance modifiers is dependent on modifier linkage.** Parameter values for the ancestral population are S = 0.1, M = 0.05, R = 0.5;  $\theta = 0$ ;  $\phi = 0$ . In Panel (A) 2 loci contribute to local adaptation. The green curves show selection on a modifier that simultaneously affects dominance of all loci, or a modifier that affects dominance of the locus tightly linked to the modifier (locus 1) or a locus that freely recombines with the modifier (locus 2). In panel (B) 5 loci contribute to local adaptation. Loci 2–5 all freely recombine with the modifier, and so selection on modifiers that affect dominance at any of these loci have the same strength of selection.

#### Ancestral levels of epistasis

While the heterozygosity of the ancestral population will determine selection on dominance modifiers, ancestral levels of linkage disequilibrium will determine selection on other modifiers of genetic architecture. In part, the ancestral population genetic equilibrium is a function of epistasis and therefore we can vary it to model how selection on the modifiers would change with an evolving genetic architecture, and not, as our modeling presupposes, that selection during the invasion is the same as when the modifier reaches intermediate frequencies.

Because we must consider a modifier of epistasis whose fitness effects are positive in one habitat and negative in the other it means that the maximum rate of spread of an epistatic modifier allele causing its bearer to have relative fitness of 1 is given by a migration rate dependent eigenvalue,  $\lambda(A)$ , where:

$$A = \begin{vmatrix} \frac{(1-m_1)}{L_1} & \frac{m_1}{L_2} \\ \frac{m_2}{L_1} & \frac{(1-m_2)}{uL_2} \end{vmatrix}$$
(8)

The theoretical maximum strength of selection for an epistasis modifier is then given by the eigenvalue of this matrix minus 1:  $\lambda(A) - 1$  (Appendix B in S1 Appendix).

As the amount of epistasis increases,  $\lambda(A) - 1$  decreases and approaches a lower asymptote when extreme epistasis effectively creates a single partially dominant locus (Fig 5). Selection







**Fig 6. Selection for modifiers depends on the strength of selection for local adaptation.** As the value of *S* is increased, the strength of selection on all types of modifier goes up (panel A). The relative advantage of epistasis and dominance modifiers, as compared with recombination modifiers, is shown in panel B. Epistasis modifiers have about a 6 fold higher invasion rate than a modifier that eliminates recombination (i.e. an inversion). Parameter values for the ancestral population are k = 3, M = 0.05, R = 0.5;  $\phi = 0$ ;  $\theta = 0$ .

https://doi.org/10.1371/journal.pgen.1010350.g006

for migration reduction increases with  $\phi$ , because the dispersal modifier can reduce a larger fraction of the load when there are fewer loci contributing to local adaptation (as in Fig 2).

Selection for additional increases in epistasis shows a complex pattern. First, as  $\phi$  increases from 0, there is an increase in selection for further increases in  $\phi$ . Selection for further increases in  $\phi$  then reaches a local maximum, and after  $\phi$  reaches about 3 then selection for further increases in  $\phi$  substantially decreases. Eventually, further selection for increased  $\phi$  is reduced to zero. When  $\phi = 0$ , selection for a complete reduction in recombination is about 1/4 as large as selection for increased  $\phi$ .

Like selection for epistasis, recombination is initially more strongly selected as  $\phi$  increases, and does not substantially decrease until  $\phi$  is around 4. Selection decreases to about 4% of the theoretical maximum, but does not decay towards 0. Once  $\phi$  is about about 4.5, selection for tight linkage supersedes selection for increased epistasis.

In contrast to the other modifier types, selection for dominance decreases as  $\phi$  increases from 0. Once moderate levels of  $\phi$  are reached, selection for tight linkage supersedes selection for dominance.

#### Ancestral local adaptation

Ancestral levels of local adaptation will also determine heterozygosity and linkage disequilibrium necessary to generate selection on the modifiers. We thus also explored varying the parameter *S* which determines the fitness cost that the most maladapted genotype experiences. This analysis further allows us consider selection on the modifier under "weak" vs "strong" selection for local adaptation.

Fig 6A shows that selection on each type of modifier increases as the fitness benefit of local adaptation goes up. While the response of selection for dispersal modifiers appears linear, the other types of modifier show non-linear functions under stronger and stronger selection. However, the relative advantage of epistasis and dominance modifiers, as compared with recombination modifiers, remains large and constant from weak to strong selection (Fig 6B).

#### Conclusions

We have presented a framework to compare the magnitude of selection between modifiers of the genetic architecture of a structured population undergoing local adaptation in two habitats with different environmental conditions. We identify four previously unreported features of selection on genetic architecture modifiers. First, selection for reduced recombination, even when the modifier is a chromosomal inversion that reduces recombination to zero, is weaker than for the other types of modifier. Second, when there is a constant total benefit of local adaptation but varying number of loci, increasing the number of loci increases selection strength for modifiers of epistasis and recombination, but decreases selection strength for dominance and migration modifiers. Third, for modifiers of dominance, the strength of selection is higher when they alter dominance at loci that the modifier itself is not linked to. Finally, even at moderate levels of epistasis in the resident population, selection for further increases in epistasis is stronger than selection for tighter linkage of the locally adapted loci. Selection on modifiers to reduce dispersal rates is complex, depending on the number of loci, epistasis and local adaptation of the ancestral population. Further study of this process would require building models that include selection for dispersal due to resource competition, source sink dynamics, or include a behavioral model for how short and long range dispersal are related.

Under migration selection balance when local populations are well mixed, then selection should act to reduce dispersal [6]. Our results for selection to decrease dispersal are in line with this in that we see consistent directional selection towards lower dispersal rates. However, we find that selection on dispersal is weaker than on other types of modifiers unless the number of local adaptation loci is low and ancestral epistasis is strong. Adding complication, it is known that other factors, such as kin selection, might play a more important role in causing selection on dispersal than levels of migration load. When local resource competition between kin is possible, relatively high levels of dispersal are often favored [48-50]. There is ample selection for dispersal to find a suitable habitat when local population density approaches the carrying capacity, and dispersal to find suitable habitat in these circumstances may lead to byproduct dispersal at different distance scales. While migration itself is responsible for the entire fitness cost that drives selection for all other modifiers here investigated, selection acts more strongly on modifiers of genetic architecture than on modifiers of the dispersal rate. This seems paradoxical, because if selection acted to reduce dispersal then selection for other genetic architecture modifiers would disappear. How can we reconcile these ideas? Part of the answer is that modifiers of dispersal only alter the rate at which individuals carrying the modifier move between habitats, and do not change the impact or rate of interaction with other genotypes that have dispersed into a local deme. Modifiers that decrease dispersal protect the individual from arriving in a habitat to which they are not adapted, but it does not protect them from the effects of mating with others who themselves dispersed. Selection for reduced migration weakens as the number of loci increase because the immediate cost to a modifier of mating with maladapted individuals decreases.

In [20] it is shown that the strength of selection for inversions under continent-island selection is mathematically related to the amount of migration load generated in the absence of an inversion and the migration load generated once an inversion is fixed. Selection on an inversion, however, is determined by the fraction of the migration load that the inversion is able to ameliorate [31]. [20] consider a scenario where there is one-way migration from a continent to an island, fitness is multiplicative (i.e. additive on log scale), and locally adapted alleles are segregating on the island at migration-selection balance. An inversion that causes locally adapted alleles to become tightly linked can spread because it causes the marginal fitness of the haplotype carrying the modifier and a full complement of locally adapted alleles to be pegged at 1, and therefore is positively selected with a magnitude of L - m, or the load minus the migration rate (Appendix B in <u>S1 Appendix</u>). Under bi-directional reciprocal dispersal we consistently find that selection to reduce the recombination rate is weaker than other changes to the genetic architecture. This can further be analyzed by considering how the modifier alters the reproductive value classes of the offspring [40]. We observe that when the modifier only partially reduces recombination then selection is quite weak.

Perhaps the most significant result of our analysis is that the strength of selection for dominance and epistasis modifiers is relatively stronger than that for recombination modifiers. This is in contrast to situations where locally adapted alleles are not yet established, linkage between the modifier and existing adaptation loci is important [27]. We further found that modifiers of dominance spread fastest when they are unlinked to the locus they modify, but are linked to another locus that also contributes to local adaptation. To better appreciate this, consider two loci, A and B, both varying for alleles that confer local adaptation between habitats. If a modifier of dominance that causes the habitat 1 allele to be dominant at locus B/b and is in tight linkage with B, then it will confer a direct advantage whenever the modifier and B are in habitat 1 but a cost when in habitat 2. If the modifier is linked to the A allele at the A/a locus, then in habitat 1 it is beneficial when the carrier is heterozygous at the B/b locus. If the modifier migrates into the habitat two, it is only costly if one of the B alleles is not locally adapted. We thus expect that genomic islands of differentiation between populations are more likely caused by selection on partially-dominant mutations that just happen to appear next to already existing locally-adapted alleles. A possible example of this process might have occurred at the selfincompatibility locus of Brassica, where a non-recombining gene cluster showing high levels of DNA sequence divergence within and between species shows a hierarchy of partial dominance in locally-adapted populations [10].

In the work of [37] an approximation is derived for the spread of a dominance modifier when there is one locus contributing to local adaptation. They find that selection on a modifier can be approximated by the weighted average of the affect on heterozygote fitness in the two demes, where the weighting is the product of heterozygote frequencies and the reproductive value of alleles in each deme. This result reflects the selective value of changing a life-history parameter is the average over changes in the product of the immediate fitness effect of the mutation with the frequency of the class experiencing the change and the reproductive value of that class. In our study, heterozygote frequency alone is not proportional to the strength of selection on the modifier (results not shown), and this could be because changes in the ecological parameters, such as the dispersal rate, also cause the reproductive value of heterozygotes to change. Further work is required to relate our selection strength estimates to the underlying components of the frequency and reproductive value of the haplotypes.

Analogous to selection on recombination modifiers, selection for assortative mating preference can occur when heterozygotes are maintained by migration-selection balance, frequency dependence, or overdominance, and the effective recombination rate between locally-adapted loci can evolve, e.g., [40, 51, 52]. Mating preference modifiers affect associations of genes without altering direct fitness, particularly for strong assortative mating, and their spread can be wholly driven by selection created by a migration load. In particular have found that when load is caused by heterozygote advantage then modifiers of dominance are selected for, and also that assortative mating is selected for [52]. When the frequency of heterozygotes is high, then selection for assortative mating can be stronger than selection for dominance. Our results here show that dominance modifiers can achieve selection coefficients of about one-third the maximum possible selection on modifiers, which means other types of modifier, such as mating preference modifiers, could also achieve substantially higher selection.

Most analyses of selection on modifiers assume that the fitness differences between alternative genotypes are small. There are two main reasons for this, one is analytical practicality and the other is the belief that the bulk of naturally segregating alleles have small effects on fitness. For the cases we studied here, both of these reasons are less applicable. Under bi-directional migration we do not have good analytical approximations for either the amount of migration load or the equilibrium genotype frequencies, so no reduction of mathematical complexity is achieved by assuming weak selection. We can still approximate selection on modifiers using a weak-load and quasi linkage equilibrium assumptions, and while these approximations perform well when numerically solving for the genotype frequency, they do not provide much additional insight (see Appendix B in S1 Appendix). Second, large differences in fitness between demes can be maintained with only a few locally-adapted loci and high dispersal rates, patterns which have been observed in natural populations, e.g. [53, 54]. Our results indicate that modifiers of epistasis or dominance consistently experience larger selection coefficients than modifiers of recombination for both weak and strong migration loads.

In sum, we find little evidence for selection for decreased recombination rates under migration load. Instead, selection should diminish migration loads by favoring modifiers of dominance and epistasis within and between locally-adapted loci. Selection on dispersal modifiers is more complex. Our modeling assumes deterministic dynamics so a venue for future study is to find out if genetic drift and demographic stochasticity qualitatively change our conclusions.

#### **Supporting information**

**S1 Appendix. Appendices.** Appendix A contains an alternative formulation for the models that uses matrix and tensor notation to represent the population genetic recursion equations. Appendix B relates the current model to models of the spread of inversions under migration selection balance.

(PDF)

#### Acknowledgments

We thank Reinhard Bürger, Joachim Hermisson, Thomas Lenormand and Denis Roze for discussion, special thanks to Nick Barton for suggesting the form for the fitness function, and two anonymous reviewers for very helpful suggestions on how to improve the manuscript.

#### **Author Contributions**

Conceptualization: Stephen R. Proulx.

Formal analysis: Stephen R. Proulx.

Writing - original draft: Stephen R. Proulx, Henrique Teotónio.

Writing - review & editing: Stephen R. Proulx, Henrique Teotónio.

#### References

- 1. Fisher RA. The genetical theory of natural selection: a complete variorum edition. Oxford University Press; 1930.
- Haldane JBS. The Effect of Variation of Fitness. The American Naturalist. 1937; 71(735):337–349. https://doi.org/10.1086/280722
- 3. Crow JE. Some possibilities for measuring selection intensity in man. Human Biology. 1958; 30:1–13.
- 4. Crow JE. Genetic loads and the cost of natural selection. In: Kojima KI, editor. Mathematical models in population genetics. Springer, Berlin; 1970. p. 128–177.
- Kondrashov AS. Deleterious mutations as an evolutionary factor. I. The advantage of recombination. Genetical Research. 1984; 44:199–217. https://doi.org/10.1017/S0016672300026392 PMID: 6510714
- Altenberg L, Liberman U, Feldman MW. Unified reduction principle for the evolution of mutation, migration, and recombination. PNAS. 2017; 114(12):E2392–E2400. <u>https://doi.org/10.1073/pnas.</u> 1619655114 PMID: 28265103

- Saastamoinen M, Bocedi G, Cote J, Legrand D, Guillaume F, Wheat CW, et al. Genetics of dispersal. Biol Rev. 2018; 93:574–599. https://doi.org/10.1111/brv.12356 PMID: 28776950
- Rockman MV, Kruglyak L. Recombinational Landscape and Population Genomics of Caenorhabditis elegans. PLOS Genetics. 2009; 5(3):1–16. <u>https://doi.org/10.1371/journal.pgen.1000419</u> PMID: 19283065
- Peñalba JV, Wolf JBW. From molecules to populations: appreciating and estimating recombination rate variation. Nature Reviews Genetics. 2020; 21(8):476–492. https://doi.org/10.1038/s41576-020-0240-1 PMID: 32472059
- Durand E, Meheust R, Soucaze M, Goubet PM, Gallina S, Poux C, et al. Dominance hierarchy arising from the evolution of a complex small RNA regulatory network. Science. 2014; 346(6214):1200–1205. https://doi.org/10.1126/science.1259442 PMID: 25477454
- 11. Billiard S, Castric V, Llaurens V. The integrative biology of genetic dominance. Biological Reviews. 2021; 96(6):2925–2942. https://doi.org/10.1111/brv.12786 PMID: 34382317
- Noble LM, Chelo I, Guzella T, Afonso B, Riccardi DD, Ammerman P, et al. Polygenicity and Epistasis Underlie Fitness-Proximal Traits in the Caenorhabditis elegans Multiparental Experimental Evolution (CeMEE) Panel. Genetics. 2017; 207(4):1663–1685. <u>https://doi.org/10.1534/genetics.117.300406</u> PMID: 29066469
- Phillips PC. Epistasis-the essential role of gene interactions in the structure and evolution of genetic systems. Nat Rev Genet. 2008; 9(11):855–67. https://doi.org/10.1038/nrg2452 PMID: 18852697
- 14. Turner TL, Hahn MW, Nuzhdin SV. Genomic islands of speciation in Anopheles gambiae. PLoS biology. 2005; 3(9):e285. https://doi.org/10.1371/journal.pbio.0030285 PMID: 16076241
- Nosil P, Funk DJ, Ortiz-Barrientos D. Divergent selection and heterogeneous genomic divergence. Molecular Ecology. 2009; 18(3):375–402. https://doi.org/10.1111/j.1365-294X.2008.03946.x PMID: 19143936
- Carneiro M, Baird SJ, Afonso S, Ramirez E, Tarroso P, Teotonio H, et al. Steep clines within a highly permeable genome across a hybrid zone between two subspecies of the European rabbit. Mol Ecol. 2013; 22(9):2511–25. https://doi.org/10.1111/mec.12272 PMID: 23530594
- Cruickshank TE, Hahn MW. Reanalysis suggests that genomic islands of speciation are due to reduced diversity, not reduced gene flow. Molecular Ecology. 2014; 23(13):3133–3157. <u>https://doi.org/10.1111/</u> mec.12796 PMID: 24845075
- Kirkpatrick M, Barton N. Chromosome Inversions, Local Adaptation and Speciation. Genetics. 2006; 434(May):419–434. https://doi.org/10.1534/genetics.105.047985 PMID: 16204214
- Bürger R, Akerman A. The effects of linkage and gene flow on local adaptation: A two-locus continent island model. TPB. 2011; 80(4):272–288. https://doi.org/10.1016/j.tpb.2011.07.002 PMID: 21801739
- Charlesworth B, Barton NH. The Spread of an Inversion with Migration. Genetics. 2018; 208(January):377–382. https://doi.org/10.1534/genetics.117.300426 PMID: 29158424
- Feder JL, Nosil P. The efficacy of divergence hitchhiking in generating genomic islands during ecological speciation. Evolution. 2010; 64:1729–1748. <u>https://doi.org/10.1111/j.1558-5646.2009.00943.x</u> PMID: 20624183
- Feder JL, Gejji R, Yeaman S, Nosil P. Establishment of new mutations under divergence and genome hitchhiking. Philosophical Transactions of the Royal Society B: Biological Sciences. 2012; 367:461– 474. https://doi.org/10.1098/rstb.2011.0256 PMID: 22201175
- Flaxman SM, Feder JL, Nosil P. Genetic hitchhiking and the dynamic buildup of genomic divergence during speciation with gene flow. Evolution. 2013; 67:2577–2591. https://doi.org/10.1111/evo.12055 PMID: 24033168
- Yeaman S. Genomic rearrangements and the evolution of clusters of locally adaptive loci. PNAS. 2013; 110:E1743–E1751. https://doi.org/10.1073/pnas.1219381110 PMID: 23610436
- Aeschbacher S, Bürger R. The Effect of Linkage on Establishment and Survival of Locally Beneficial Mutations. Genetics. 2014; 197(May):317–336. https://doi.org/10.1534/genetics.114.163477 PMID: 24610861
- Yeaman S, Aeschbacher S, Bürger R. The evolution of genomic islands by increased establishment probability of linked alleles. Moleculary Ecology. 2016; 25:2542–2558. <u>https://doi.org/10.1111/mec. 13611</u> PMID: 27206531
- Pontz M, Bürger R. The effects of epistasis and linkage on the invasion of locally beneficial mutations and the evolution of genomic islands. Theoretical Population Biology. 2022; 144:49–69. https://doi.org/ 10.1016/j.tpb.2022.01.004 PMID: 35124094
- Bank C, Bürger R, Hermisson J. The Limits to Parapatric Speciation: Dobzhansky–Muller Incompatibilities in a Continent–Island Model. Genetics. 2012; 191(3):845–863. https://doi.org/10.1534/genetics. 111.137513 PMID: 22542972

- Blanckaert A, Hermisson J. The Limits to Parapatric Speciation II: Strengthening a Preexisting Genetic Barrier to Gene Flow in Parapatry. Genetics. 2018; 209(1):241–254. https://doi.org/10.1534/genetics. 117.300652 PMID: 29496748
- Akerman A, Bürger R. The consequences of gene flow for local adaptation and differentiation: a twolocus two-deme model. Mathematical Biology. 2014; 68:1135–1198. <u>https://doi.org/10.1007/s00285-013-0660-z PMID: 23532261</u>
- Proulx SR, Phillips PC. The opportunity for canalization and the evolution of genetic networks. Am Nat. 2005; 165(2):147–62. https://doi.org/10.1086/426873 PMID: 15729647
- **32.** Fisher RA. The possible modification of the response of the wild type to recurrent mutations. Am Nat. 1928; 62:115–126. https://doi.org/10.1086/280193
- 33. Wright SG. Fisher's theory of dominance. Am Nat. 1929; 63:274–279. https://doi.org/10.1086/280260
- Feldman MW, Karlin S. The evolution of dominance: a direct approach through the theory of linkage and selection. Theor Popul Biol. 1971; 2(4):482–92. https://doi.org/10.1016/0040-5809(71)90036-0 PMID: 5170720
- Karlin S, McGregor J. Towards a theory of the evolution of modifier genes. Theor Popul Biol. 1974; 5 (1):59–103. https://doi.org/10.1016/0040-5809(74)90052-5 PMID: 4818408
- Wagner GP, Bürger R. On the evolution of dominance modifiers II: a non-equilibrium approach to the evolution of genetic systems. J Theor Biol. 1985; 113(3):475–500. <u>https://doi.org/10.1016/S0022-5193</u> (85)80034-5 PMID: 3999784
- Otto SP, Bourget D. Balanced polymorphisms and the evolution of dominance. Am Nat. 1999; 153 (6):561–574. https://doi.org/10.1086/303204 PMID: 29585648
- Van Dooren TJM. The Evolutionary Ecology of Dominance-Recessivity. JTB. 1999; 198:519–532. https://doi.org/10.1006/jtbi.1999.0929 PMID: 10373352
- Phillips PC, Otto SP, Whitlock M. In: Wolf JB, Brodie ED III, Wade MJ, editors. Beyond the Average: The Evolutionary Importance of Gene Interactions and Variability of Epistatic Effects. New York: Oxford University Press; 2000.
- Proulx SR, Servedio MR. Dissecting selection on female mating preferences during secondary contact. Evolution. 2009; 63:2031–2046. https://doi.org/10.1111/j.1558-5646.2009.00710.x PMID: 19473397
- Szep E, Sachdeva H, Barton NH. Polygenic local adaptation in metapopulations: A stochastic eco-evolutionary model. Evolution. 2021; 75(5):1030–1045. <u>https://doi.org/10.1111/evo.14210</u> PMID: 33742441
- Kirkpatrick M, Johnson T, Barton N. General Models of Multilocus Evolution. Genetics. 2002; 161 (4):1727–1750. https://doi.org/10.1093/genetics/161.4.1727 PMID: 12196414
- Kacser H, Burns JA. The molecular basis of dominance. Genetics. 1981; 97(3-4):639–666. <u>https://doi.org/10.1093/genetics/97.3-4.639 PMID: 7297851</u>
- 44. Proulx, S, Teotonio, H. The relative strength of selection on modifiers of genetic architecture under migration load, Dryad, Dataset 2022
- Chelo IM, Nédli J, Gordo I, Teotónio H. An experimental test on the probability of extinction of new genetic variants. Nature communications. 2013; 4(1):1–8. https://doi.org/10.1038/ncomms3417 PMID: 24030070
- 46. Rice SH. A general population genetic theory for the evolution of developmental interactions. Proceedings of the National Academy of Sciences. 2002; 99(24):15518–15523. <u>https://doi.org/10.1073/pnas.</u> 202620999 PMID: 12438697
- 47. Grieshop K, Ho EKH, Kasimatis KR. arXiv preprint: Dominance reversals, antagonistic pleiotropy, and the maintenance of genetic variation; 2021.
- Hamilton WD, M MR. Dispersal in stable habitats. Nature. 1977; 269:578–581. <u>https://doi.org/10.1038/</u> 269578a0
- **49.** Ronce O. How does it feel to be like a rolling stone? Ten questions about dispersal evolution. Annual review of ecology, evolution, and systematics. 2007; 38:231–253. https://doi.org/10.1146/annurev. ecolsys.38.091206.095611
- Billiard S, Lenormand T. Evolution of migration under kin selection and local adaptation. Evolution. 2005; 59(1):13–23. https://doi.org/10.1554/04-232 PMID: 15792223
- Otto SP, Servedio MR, Nuismer SL. Frequency-Dependent Selection and the Evolution of Assortative Mating. Genetics. 2008; 179(4):2091–2112. https://doi.org/10.1534/genetics.107.084418 PMID: 18660541
- Durinx M, Van Dooren TJM. Assortative mate choice and dominance modification: alternative ways of removing heterozygote disadvantage. Evolution. 2009; 63(2):334–352. https://doi.org/10.1111/j.1558-5646.2008.00578.x PMID: 19054051

- Hoekstra HE, Krenz JG, Nachman MW. Local adaptation in the rock pocket mouse (Chaetodipus intermedius): natural selection and phylogenetic history of populations. Heredity. 2005; 94(2):217–228. https://doi.org/10.1038/sj.hdy.6800600 PMID: 15523507
- 54. Hall MC, Willis JH. Divergent selection on flowering time contributes to local adaptation in Mimulus guttatus populations. Evolution. 2006; 60(12):2466–2477. https://doi.org/10.1111/j.0014-3820.2006. tb01882.x PMID: 17263109